Author's response to reviews

Title: Multiple micronutrient supplementation improves vitamin B12 and folate concentrations of HIV infected children in Uganda: a randomized controlled trial

Authors:

Grace Ndeezi Dr (gracendeezi@yahoo.com)
James K Tumwine Prof (kabaleimc@gmail.com)
Christopher M Ndugwa Prof (cmndugwa@yahoo.co.uk)
Bjørn J Bolann Prof (Bjorn.Bolann@med.uib.no)
Thorkild Tylleskår Prof (Thorkild.Tylleskar@cih.uib.no)

Version: 3 Date: 27 April 2011

Author's response to reviews:

Dear Independent Editor,

We have revised the Authors' list to match what is in the submission system, addressed the minor issues and improved on the language and punctuation. Thank you very much for your suggestions.

Best regards,

Grace Ndeezi